Author's response to reviews

Title: Psychotropic medication in the French child and adolescent population: prevalence estimation from health insurance data and national self-report survey data

Authors:

eric acquaviva (dracquaviva@aol.com)
stephane legleye (stleg@ofdt.fr)
guy robert auleley (guy-robert.auleley@le-rsi.fr)
jean deligne (jean.deligne@cmr-npc.canam.fr)
didier carel (dcarel@urcam.org)
bruno falissard (falissard-b@wanadoo.fr)

Version: 3 Date: 21 October 2009

Author's response to reviews:

Dear Editor,

In response to the comments made by the Associate Editor, we have made revisions to our manuscript “Psychotropic medication in the French child and adolescent population: prevalence estimation from social security data and results from a national self-report study”.

First, we have changed “benzodiazepin” to “benzodiazepine”.

Moreover, in the last paragraph of the conclusion section, we have changed “external disorders” to “certain psychiatric disorders”

Hoping these revisions meet with your approval.

Yours sincerely,

Eric Acquaviva, MD

Service de Psychopathologie
de l’ Enfant et de l’ Adolescent
48 Boulevard Serurier,
75019 Paris, France
Phone : +33140032263
Fax : +33140032297
e-mail : eacqua7129@aol.com